Author's response to reviews

Title: The context and practice of handwashing among new mothers in Serang, Indonesia: a formative study

Authors:

Katie Greenland (katie.greenland@lshtm.ac.uk)
Endang Iradati (eiradati@jhpiego.net)
Abigail Ati (abby.ati@gmail.com)
Yanti Yulianti Maskoen (butimaskoen@yahoo.com)
Robert Aunger (Robert.aunger@lshtm.ac.uk)

Version: 5 Date: 20 August 2013

Author's response to reviews: see over
Reviewer's report
Title: The context and practice of handwashing among new mothers in Serang, Indonesia: a formative study
Version: 4 Date: 30 July 2013
Reviewer: Robert Dreibelbis
Reviewer's report:
Minor essential revisions:
1) P. 3: There seems to be something going on with the references in this paragraph. Specifically: "largely stem from reductions in diarrhoeal diseases [2, 3], RI Ejemot, JE Ehiri, MM Meremikwu and JA Critchley [4], a common cause of morbidity:
Otherwise, I believe that all previous comments have been addressed!
Level of interest: An article whose findings are important to those with closely related research interests
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.
Declaration of competing interests:
I declare that I have no competing interest.

Response to Reviewer

Apologies, sometimes Endnote has a mind of its own and didn’t update this reference correctly. I am pleased that you took time to notice the reference issue. This has now been resolved.